Accelerator Status Report

Run Schedule: Aug 21, 2009 to Dec 23, 2009
Sept. 16, 2009

Injector

· During Saturday swing shift (Sept. 12) after some injector beam line vacuum valve closed and RF 0L03 tripped, beam would not transport beyond BPM0L06. After confirming all the injector parameters were where they were supposed to be Ops suspected that a vacuum valve was not fully open even the control screen indicated all the valves were open. After cycling the valves many times beam transported.

· Laser spot was moved yesterday, Sept. 15. 

· Wien angle is 17 degrees.
· Injector was optimized to stabilize Hall B current.

· Wien angle is 14 degrees.

Accelerator

· DOG2E and DOG3E tripped off several times and EES fixed them.
· RF separators had cooling problems and kept tripping. They were acid flushed yesterday. 
· We have tightened up ion chamber trip levels in Hall A and Hall C. We may have to relax the trip levels if we have more trips but make sure beam transports well when adjusting them.
· Overall the accelerator was running well with high current to Hall A.

· PSS dropped the NL to restricted access due to a failed UPS. It dropped Hall C to restricted access a couple of times. 

· Iocse20 went unresolved a couple of times. Turn off the orbit lock if the ioc is unresolved because the orbit lock may steer beam away.

· Ion chamber setpoints: We may need a better way to do it so we can have better protection. 

· Doglegs 4 and 8 are out of room. 

· RF and magnets issues.

· RF 1L04 commissioning started, about ten hours each week.
· Sweep procedure has been updated.

Hall A---E05-109; pass 3; energy 3.49 GeV
· With optics on-call support many hours were spent to lower and stabilize Hall A Compton counts. More time is needed to troubleshoot the problem.
· Hall A Compton ion chamber sometimes trips off even with on beam in the hall. We may have to raise the trip level a bit. We are having problems raising the Compton trip level. It would not go higher than 1000 and Lahti is looking into it.
· Hall A Compton rates were high. 1C20 quad was adjusted to tighten the beam, which lowered the rates.
Hall B ---Eg1-DVCS; pass 5; energy 5.764 GeV
· Monday Hall B reported that we sent too much beam to their during the beam checkout after a down. Hall B has a polarized target. The polarization of its target could be spoiled by current higher than 10nA. If we need to send tune beam to the hall line we can let the hall move out its target or send beam to its Tagger dump. 
· Hall B orbit lock was fixed and has been working.
· Hall B current was not stable. Reza optimized the injector which stabilized it.

· Orbit lock still has problems.

Hall C---HKS; pass 2; energy 2,344 GeV
· Hall C SLI indicates the energy spread higher than required (5E-5). 
· Hall C beam sizes are larger than required (50um x 100um).

· After MDW3H04AH power supply was swap beam was steered to the YAG viewer.

· SLI needs does not work yet. EES just make the slit motor work.

· Due to iocse20 the hall detector may have been damaged due to the mis-steered beam.

· Hall C still has harp issues.

http://opweb.acc.jlab.org/CSUEApps/elog/elog.php
